Artificial Intelligence 2A, 2004–2005 PROVISIONAL TIMETABLE

AI2A consists of two Modules, each consisting of two Tasks:

Module 1 (Bob Fisher)

Task 1: Using constraints to solve problems. 7 lectures.

Task 2: Finding solutions by other means. 7 lectures.

Module 2 (Bonnie Webber)

Task 3: Pattern matching in reasoning. 7 lectures.

Task 4: Pattern matching in analysis. 7 lectures.

Lecture and tutorial schedule

Lectures are at 4:10 pm in Lecture Theatre 3, Appleton Tower (Mondays and Thursdays) and Room G8, 1 George Square (Tuesdays).

Week w/c Monday Tuesday Thursday Tutorial topic 1 20 Sept holiday Intro (jrl) Task 1 – 2 27 Sept Task 1 Task 1 Task 1 Task 1 3 4 Oct Task 1 Task 1 Task 1 Task 1 4 11 Oct Task 2 Task 2 Task 2 Task 2 5 18 Oct Task 2 Task 2 Task 2 Task 2 6 25 Oct break break Task 2 – 7 1 Nov Task 3 Task 3 Task 3 Task 2 8 8 Nov Task 3 Task 3 Task 3 Task 3 9 15 Nov Task 3 Task 4 Task 4 Task 3 10 22 Nov Task 4 Task 4 Task 4 Task 4 11 29 Nov Task 4 Task 4 – Task 4

Monday 20 September (the first day of our new semester system!) is a public holiday. There is a mid-semester break from Sat 23 Oct to Tues 26 Oct.

Practical schedule

Practical Set by Submission deadline A1 Bob Fisher Friday 15 Oct (Week 4) A2 Bob Fisher Friday 5 Nov (Week 7) A3 Bonnie Webber Friday 19 Nov (Week 9) A4 Bonnie Webber Friday 3 Dec (Week 11)

Submission deadlines for practicals will be at 4pm on Fridays.